Author's response to reviews

Title: No association between the aluminium content of trabecular bone and bone density, mass or size of the proximal femur in elderly men and women

Authors:

Hans-Olov Hellstrom (hans.lov.hellstrom@akademiska.se)
Bengt Mjoberg (bengt.mij@telia.com)
Hans Mallmin (hans.mallmin@akademiska.se)
Karl Michaelsson (karl.michaelsson@akademiska.se)

Version: 2 Date: 23 March 2006

Author's response to reviews: see over
Dear Editor(s)

Please consider for publication in BMC Musculoskeletal Disorders the enclosed manuscript (No association between the aluminium content of trabecular bone and bone density, mass or size of the proximal femur in elderly men and women). This original manuscript has not been published elsewhere.

Yours sincerely

Hans-Olov Hellström, MD
Department of orthopaedics
Uppsala University hospital
S-751 85 Uppsala
+46 18 611 44 79
hans.lov.hellstrom@akademiska.se